Author's response to reviews

Title: Lineage relationship of prostate cancer cell types based on gene expression

Authors:

Laura E Pascal (pascalle@upmc.edu)
Ricardo ZN Vencio (rvencio@rge.fmrp.usp.br)
Robert L Vessella (vessella@uw.edu)
Carol B Ware (cware@uw.edu)
Eneida F Vencio (vencio56@hotmail.com)
Gareth Denyer (gareth@usyd.edu.au)
Alvin Y Liu (aliu@uw.edu)

Version: 5 Date: 10 May 2011

Author's response to reviews: see over
May 10, 2011

RE: MS 5688587314644422

Major Revisions

Title page: We have included author affiliations.

Abstract: We have revised the Abstract structure accordingly.

Competing interests: We have included a competing interests statement between the Conclusions and Authors’ contributions.

References: We have fixed the reference 29 citation within the main text and references.

Tables: We have added the tables to the manuscript as suggested.

Figures: We have revised the figures to not include the title or figure number.

Typography: We have checked for typographical errors.

We sincerely appreciate the thorough review of our manuscript. The constructive suggestions helped us to greatly improve and focus our work for this revised manuscript.

Sincerely,

Hunter Wessells, MD -
Chairman

Faculty

Paul D. Axford, MD
Ramasesamy Bakthavatsalam, MD
James A. Bassuk, PhD
Richard E. Berger, MD
Kanchan A. Chitaley, PhD
Eva Corey, PhD
Bruce L. Dalkin, MD
William J. Ellis, MD
John L. Gore, MD
Richard W. Grady, MD
Jonathan D. Harper, MD
Celestia S. Higano, MD
Byron D. Joyner, MD
Martin A. Koyle, MD
John N. Krieger, MD
Paul H. Lange, MD
Thomas S. Lendvay, MD
Daniel W. Lin, MD
Alvin Y. Liu, PhD
Jane L. Miller, MD
Colm M. Morrissey, PhD
Michael P. Porter, MD
Donald E. Riley, PhD
Margarett Shnorhavorian, MD
Robert L. Vessella, PhD
Bryan B. Voelzke, MD
Thomas J. Walsh, MD
Jonathan L. Wright, MD
Claire C. Yang, MD
Laura E. Pascal, Ph.D. 

Alvin Y. Liu, Ph.D. 
Research Associate Professor 
aliu@uw.edu 
(206) 221-0603